Author's response to reviews

**Title:** Elevated Platelet Derived Growth Factor-BB concentrations in premature neonates who develop chronic lung Disease

**Authors:**
- Kim G Adcock (kadcock@pharmacy.umsmed.edu)
- Jeremy Martin (jmarti9@lsuhsc.edu)
- John Loggins (jloggi@lsuhsc.edu)
- Thomas E Kruger (dkruger1@kc.rr.net)
- Dr R John Baier (jbaier@lsuhsc.edu)

**Version:** 5 **Date:** 14 Jun 2004

All of the formatting changes have been made to the revised manuscript. Thanks John Baier MD